Author's response to reviews

Title: Maternal outcome of delivery among adolescent girls at the Yaounde Central Hospital

Authors:

Florent Ymele Fouelifack (yfouelifack@gmail.com)
Philip Njotang Nana (nananjotangphilip@yahoo.com)
Theodore Yangsi Tameh (theotameh@yahoo.co.uk)
Ngole Eta Mbong (mbongeta@yahoo.fr)
Hortence Jeanne Fouedjio (fouedjiojeanne@yahoo.fr)
Robinson Enow Mbu (rembu2000@yahoo.com)

Version: 2 Date: 9 May 2013

Author's response to reviews:

Dear Sir/Madam

We hereby submit the reviewed manuscript of the study some and I proposed for publication in your journal with the following rectifications:

1. The English language has been improved as well as the write-up as a whole
2. The abstract has been shortened to 250 words and the referencing reviewed
3. We have also mentioned the ethical considerations we used to assure confidentiality

We remain available for any further comments you will have about the manuscript